
Daily Status Fri 2012-Oct-26
Weather:
Got about 3.0" of snow making for a beautiful day.  Not much of a wind and the house stayed warm.

Status:
All systems look great.  Fans seemed to take this snow a lot better than the first round.  All optical sensors seem to have low signals so I will take a gander 
at their condition this morning.

Morning Update:  Cleaned all Kryptons, Licors and radiometers.  After cleaning 10m Krypton, no change to signal level.  I thought I cleared off all snow and 
cleaning water...we may want to keep an eye on this sensor.  The reason I could do the Krytpon cleaning is RAL has a snow depth scientist doing 
measurements today, so he was around to dial 9-1...

I also knocked off snow on some of the solar panels.  I will not do that for the rest of the sites.  The sun is trying to pop it's head through, so I will let Mother 
Nature do her thing.

ToDo:
Clean all optic sensors (Licor, Krypton and radiation).  Krypton cleaning will depend on a sighter back at the compound.  If no sighter, then no cleaning 
Kryptons.

I will also switch out PockeTecs before leaving.  
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